Author's response to reviews

Title: Early Palliative Care - Health services research and implementation of sustainable changes: the study protocol of the EVI project

Authors:

Cornelia Meffert (cornelia.meffert@uniklinik-freiburg.de)
Jan Gaertner (jan.gaertner@uniklinik-freiburg.de)
Katharina Seibel (katharina.seibel@uniklinik-freiburg.de)
Karin Jors (karin.jors@uniklinik-freiburg.de)
Hubert Bardenheuer (hubert.bardenheuer@med.uni-heidelberg.de)
Dieter Buchheidt (dieter.buchheidt@umm.de)
Regine Mayer-Steinacker (Regine.Mayer-Steinacker@uniklinik-ulm.de)
Maren Viehrig (maren.viehrig@med.uni-tuebingen.de)
Christina Paul (paul@tropenklinik.de)
Stephanie Stock (stephanie.stock@uk-koeln.de)
Carola Xander (carola.xander@uniklinik-freiburg.de)
Gerhild Becker (gerhild.becker@uniklinik-freiburg.de)

Version: 7 Date: 26 November 2014

Author's response to reviews:

Dear Annie Lyn Bravo,

Thank you for your answer to our submission. We resubmit our files now with the changes requested:

1. We included the registration date next to the TRN at the end of the abstract.

2. On page 5, line 44, we included the following sentence: “It is funded by the Robert Bosch Foundation who has formally peer reviewed our study protocol before assigning the grant.”

We would be grateful for acceptance of our article and are looking forward to hearing from you.

Best regards

Cornelia Meffert